Author's response to reviews

Title: Elderly surgical patients and hospital bed occupancy

Authors:

Muhammad umair Majeed (umairmajeed188@gmail.com)
Dean Thomas Williams (deantwilliams@aol.com)
Rachel Pollock (rachelpollock82@yahoo.co.uk)
Farhat Amir (farhatamir@yahoo.co.in)
M Liam (m.liam@yahoo.co.uk)
K S Foong (ksfoong@yahoo.co.uk)
C J Whitaker (c.j.whitaker@bangor.ac.uk)

Version: 4 Date: 12 June 2012

Author's response to reviews:

Dear Sir/Madam,

I have made following changes to the manuscript.

1) I have included the abstract in the original manuscript.

2) The data included in the study was openly available to the vascular surgery firm in the form of patient case notes and this has been stated clearly at the end of manuscript.

Regards